Author's response to reviews

Title: Prevalence of peripheral arterial disease in subjects with moderate cardiovascular risk: Italian results from the PANDORA study

Authors:

Guido Sanna (metis@fimmg.org)
Donatella Alesso (doalesso@tin.it)
Malek Mediati (malekmediati@libero.it)
Claudio Cimminiello (claudio.cimminiello@aovimercate.org)
Claudio Borghi (claudio.borghi@unibo.it)
Amalia Lucia Fazzari (amyfazzari@tin.it)
Mario Mangrella (Mario.Mangrella@astrazeneca.com)

Version: 5 Date: 20 September 2011

Author's response to reviews:

Dear Editorial Team,

The following changes have been made to the manuscript:

1) All Highlighting/tracking has now been removed from the manuscript.

2) Structure: Acknowledgements has now been placed after Authors' Contributions, Competing Interests after the Conclusions and the Authors' Contributions directly after.

3) The manuscript has also been checked for typographical errors.

4) The authors are the PANDORA group, therefore "the PANDORA group" should not be treated as a separate additional author as requested by editorial staff.